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Instrument List:

Green

Discussion: 

All six 2DVDs should now be operat ing and collect ing data. Major opt ical adjustments were
performed today for SN36. The problem with SN38 seems to have been corrected. Data
collect ion for SN25 cannot be verif ied as it  has not been visited since deployment.

Each unit  st ill requires calibrat ion. This will be at tempted for a couple units on Monday, weather
dependent.

Progress has been made on the cellular modems. Data f low and remote communicat ions
should be possible next week.


